Notes of the RICH1 Phone meeting 04/08/2006
People Present:
Bristol:  Richard
CERN:  Dave, Fabio, Richard P.
IC:         Bill, Gabriel, Trevor
Oxford:  Rohan
RAL:     Mike, Peter
Apologies for absence:
CERN: Geoff, Kenneth, Nick
Discussed Agenda Topics:
1. Gas Enclosure 
Last Wednesday the GE was re-adjusted and surveyed. The results from the survey show that the GE has been adjusted as wanted. The opening on the front face is well aligned in x-y with a small lack in parallelism ~0.5 mm. Overall the alignment of the GE in x is very good. At the EW it will be ~1-2 mm high in y with respect to beampipe, but there is enough tolerance available for the beampipe.  There is no need for further adjustments.  All attachment points for the optical components have been surveyed and are available on EDMS. One door O-ring was damaged and will need to be replaced.  
2. VELO Seal
The VELO vacuum problems have been resolved. All external leaks have been fixed. The reason for not obtaining the same level of vacuum as at fabrication time is thought to be due to some trapped pockets of air.  The foreseen installation dates are: remove the VELO flange the 14/08/06 and install Seal to VELO the 15/08/06. All installation tooling is at CERN and is on the platform in the pit and manpower is available.

3. Exit Window

Moulds arrived last Friday and the test rig will be ready next week. The installation tooling drawings have been modified to take in account the TT rails. The installation is foreseen for Sept06.  The drawings of the boss for the EW were given to the manufacturer. No problems are foreseen for its manufacture. A quote for the boss is expected next week.
4. Beampipe

The installation rails will be placed inside the magnet. They will be installed during the next two weeks. The upper flat mirror support rail will be installed before the beampipe goes in. It is expected that the installation will begin from August 17th.

5. Spherical Mirrors
The C4F10 tank was opened on the 27/07/06. Spherical mirror sample 1 was measured showing no change in the D0, R, and reflectivity. Flat sample A surface roughness was measured and it is unchanged. The mechanics tests on the flat sample were not done because it would have required another week. The D0 of the demonstrator mirror improved from ~1.5mm to ~0.8mm. This is probably due to overstressing the mirror on the support when it was first measured. The samples (flat sample A, spherical mirror 1, and the demonstrator mirror) were put back in the C4F10 tank on the 3/08/06. 
A surface analysis was carried out on the glass samples that CMA sent to us.  The result is that the MgF2 layer in the sample with 20nm MgF2 was too thin to properly protect the aluminium, and that there is a layer of oxide seen between the aluminium and the MgF2 on the other sample. Andre will talk to CMA next Monday to discuss the coating. CMA should send us some other samples coated and uncoated (action Fabio). 
6. Flat Mirrors
A trial fit of a RICH1 mirror on the two upper mirror support plates was done in building 156 and all fits well.  The lower plates are being machined at IC.  The lower rail is in building 156. The upper rail has to be installed before the installation of the beampipe (safety requirement). The alignment setup will be prepared in building 156 (action Fabio).

7. HPD Assembly
The concept for the top and bottom HPD slide system was sent around by Rohan. The design of the photon funnel has been sent to Syracuse.  The design of the cooling plates needs to be ready by Sept06. There is no problem foreseen with the delivery of the copper plates and tubes.
8.  Quartz Windows
The order is out. The delivery date is 28 Sept06 at Prevessin; all 4 plates.
9.  Magnetic Distortions

The photon funnel is being designed to provide the Syracuse MDMS system with a nominal working volume. Ray Mountain is assessing whether this is adequate. If it proves insufficient or incompatible then the backup system (projector) can be used (~3 months time required).

10. Next meeting
Friday 18th August 2006 at 10:00(GMT) 11:00(CET). 
